\begin{table}%t2
%\centering

\caption{\label{tab:refs}References.}
\small  \begin{tabular}{lll}
           \hline\hline
	    & Observations & Model \\
	   \hline
	   NGC~4501 & Vollmer et~al. (\cite{Vollmer2008}) & Vollmer et~al. (\cite{Vollmer2008}) \\
	   NGC~4330 & Chung et~al. (\cite{Chung2007}) & Vollmer et~al. (in prep.) \\
	    & Abramson et~al. (in prep.) & \\
	   NGC~4438 & Vollmer et~al. (\cite{Vollmer2005}) & Vollmer et~al. (\cite{Vollmer2005}) \\
	   NGC~4522 & Kenney et~al. (\cite{Kenney2004}) & Vollmer et~al. (\cite{Vollmer2006}) \\
	            & Vollmer et~al. (\cite{Vollmer2004}b) & \\
	   NGC~4388 & Vollmer \& Huchtmeier (\cite{Vollmer2003}) & Vollmer \& Huchtmeier (2003) \\
	            & Oosterloo \& van Gorkom (\cite{Oosterloo2005}) & \\
	   NGC~4569 & Vollmer et~al. (\cite{Vollmer2004}a) & Vollmer et~al. (\cite{Vollmer2004}a) \\
	   \hline
        \end{tabular}
\end{table}